Author's response to reviews

Title: Pulmonary fungus ball caused by Penicillium capsulatum in a patient with type 2 diabetes: a case report and literature review

Authors:

Min Chen (chenmin96112332@yahoo.com.cn)
Jos Houbraken (j.houbraken@cbs.knaw.nl)
Weihua Pan (panweihua@medmail.com.cn)
Hao Peng (ncxq19811031@yahoo.com.cn)
Lihui Wu (wullhui72@126.com)
Deqiang Xu (dgxu@fudan.edu.cn)
Yiping Xiao (windyxiao1979@gmail.com)
Zhilong Wang (wangzn007@163.com)
Wanqing Liao (liaowanqing@sohu.com)
Chao Zhang (zhangchao324@qq.com)

Version: 5 Date: 23 August 2013

Author's response to reviews: see over
Dear Editors,

Thank you very much for your advice on our case report (MS 9615314579494818).
We have revised the manuscript in light of the editor’s comment, provided a point-by-point response to the concerns, and would like to re-submit it for your consideration. The amendments are highlighted in blue in the revised manuscript, and all authors have checked and approved the revised manuscript.
We hope that the revised manuscript could be acceptable, and we are looking forward to hearing from you soon.

Kind regards,

Yours sincerely,

Wanqin Liao, MD
Academician of Chinese Academy of Engineering.
Shanghai Key Laboratory of Molecular Medical Mycology, Shanghai Changzheng Hospital, Second Military Medical University, Shanghai, China.
Phone: 86-021-81885493
Email: liaowanqing@sohu.com
**Point-by-point response to the editor:**

Fristly, we would like to express our sincere thanks to the editors for the constructive, positive and professional comments on our manuscript.

1. Line 148. Despite that the clinical strain can be considered resistant to......
   
   **Our answer:** Yes, we have modified the sentence as “Despite the clinical strain can be considered resistant to ......”

2. Line 150. We believe that combined antifungal drug (caspofungin)......
   
   **Our answer:** Yes, we have modified the sentence as “We believe that combined antifungal drug (caspofungin + fluconazole)”.

3. Line 152. I suggest to remove the final part of the sentence, and it also revealed that fungal infections due to resistant isolates still respond -60% of the time, because one single case do not allow such a firm and generalized conclusion.

   **Our answer:** We have deleted the final part of the sentence. Indeed, one single case do not allow such a firm and generalized conclusion.